Author’s response to reviews

Title: Impact of maternal HIV-1 viremia on lymphocyte subsets among HIV-exposed uninfected infants: Protective mechanism or immunodeficiency?

Authors:

Fatima Kakkar (fatima.kakkar@umontreal.ca)
Valerie Lamarre (Valerie_Lamarre-pediatrie@ssss.gouv.qc.ca)
Thierry Ducruet (thierry.ducruet@gmail.com)
Marc Boucher (boucherhsj@aol.com)
Silvie Valois (cmis@recherche-ste-justine.qc.ca)
Hugo Soudeyns (hugo.soudeyns@recherche-ste-justine.qc.ca)
Normand Lapointe (normand_lapointe@ssss.gouv.qc.ca)

Version: 2
Date: 12 February 2014

Author’s response to reviews: see over
To the editors,

Enclosed is a revised version of the manuscript “Impact of maternal HIV-1 viremia on lymphocyte subsets among HIV-exposed uninfected infants: Protective mechanism or immunodeficiency?” with the specific changes detailed below:

Page 1, lines 16-23: Email addresses provided for all authors on the manuscript.

Page 12, line 19: Declaration of no competing interests provided.

Please let me know if any further revisions are necessary.
Thank you for your consideration,

Fatima Kakkar
Department of Pediatrics & Infectious Diseases
CHU Sainte-Justine
Montreal, Quebec Canada
Office: 514-345-4931 ext: 6885
Email: fatima.kakkar@umontreal.ca